Author’s response to reviews

Title: Relationships between burnout, turnover intention, job satisfaction, job demands and job resources for mental health personnel in an Australian mental health service

Authors:

Justin Scanlan (justin.scanlan@sydney.edu.au)
Megan Still (megan.still@health.nsw.gov.au)

Version: 2 Date: 09 Dec 2018

Author’s response to reviews:

Thank you for the prompt review of the revised manuscript. We have made the few minor changes that were requested:

(1) Email addresses for each author have been added

(2) "Introduction” changed to "Background"

(3) Manuscript has been proofread and minor spelling errors have been corrected.

Many thanks

Justin Scanlan and Megan Still